Author's response to reviews

Title: Delayed circadian phase is linked to glutamatergic functions in young people with affective disorders: A proton magnetic resonance spectroscopy study

Authors:

Sharon L Naismith (sharon.naismith@sydney.edu.au)
Jim Lagopoulos (jim.lagopoulos@sydney.edu.au)
Daniel F Hermens (daniel.hermens@sydney.edu.au)
Django White (django.white@sydney.edu.au)
Rebecca Robillard (rebecca.robillard@sydney.edu.au)
Elizabeth M Scott (e.scott@sydney.edu.au)
Ian B Hickie (ian.hickie@sydney.edu.au)

Version: 2
Date: 28 May 2014

Author's response to reviews:

Thank you for the opportunity to resubmit this manuscript with the line numbers added.

We hope you now find this manuscript appropriate for publication in BMC Psychiatry.

Kind Regards,
A/Prof Sharon Naismith